CrossMark

Citation: Honda JR, Hasan NA, Davidson RM, Williams MD, Epperson LE, Reynolds PR, et al. (2016) Environmental Nontuberculous Mycobacteria in the Hawaiian Islands. PLoS Negl Trop Dis 10(10): e0005068. doi:10.1371/journal. pntd.0005068

Editor: Mathieu Picardeau, Institut Pasteur, FRANCE

Received: April 27, 2016

Accepted: September 23, 2016

Published: October 25, 2016

Copyright: © 2016 Honda et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Partial rpoB gene sequences from 166 Hawaiian Island NTM isolates (KU128541-KU128706) and 33 M. chimaera isolates from the continental U.S. (KU144826-KU144858) were deposited in the GenBank nucleotide database. The GenBank accession numbers for type strain and representative isolate rpoB gene sequences of M. porcinum, M. abscessus, M. chelonae, and M. chimaera from NCBI are listed in S1 Table.

Funding: JRH acknowledges the University of Colorado AMC Division of Pulmonary Medicine and

RESEARCH ARTICLE

Environmental Nontuberculous Mycobacteria in the Hawaiian Islands

Jennifer R. Honda^{1,2,3}*, Nabeeh A. Hasan³, Rebecca M. Davidson³, Myra D. Williams⁴, L. Elaine Epperson³, Paul R. Reynolds³, Terry Smith⁴, Elena lakhiaeva⁴, Matthew J. Bankowski^{5,6}, Richard J. Wallace, Jr.⁷, Edward D. Chan^{1,2,3}, Joseph O. Falkinham, III⁴, Michael Strong^{1,3}

1 Division of Pulmonary Sciences and Critical Care Medicine, University of Colorado Denver, Anschutz Medical Campus, Aurora, Colorado, United States Of America, 2 Denver Veterans Affairs Medical Center, Denver, Colorado, United States Of America, 3 National Jewish Health, Denver, Colorado, United States Of America, 4 Virginia Tech, Blacksburg, Virginia, United States Of America, 5 Diagnostic Laboratory Services Inc., Aiea, Hawai'i, United States Of America, 6 Departments of Pathology and Tropical Medicine, Medical Microbiology and Pharmacology, John A. Burns School of Medicine, the University of Hawai'i at Manoa, Honolulu, Hawai'i, United States Of America, 7 University of Texas Health Science Center, Tyler, Texas, United States Of America

These authors contributed equally to this work.

* Jennifer.Honda@ucdenver.edu, hondaJ@njhealth.org)

Abstract

Lung disease caused by nontuberculous mycobacteria (NTM) is an emerging infectious disease of global significance. Epidemiologic studies have shown the Hawaiian Islands have the highest prevalence of NTM lung infections in the United States. However, potential environmental reservoirs and species diversity have not been characterized. In this cross-sectional study, we describe molecular and phylogenetic comparisons of NTM isolated from 172 household plumbing biofilms and soil samples from 62 non-patient households and 15 respiratory specimens. Although non-uniform geographic sampling and availability of patient information were limitations, *Mycobacterium chimaera* was found to be the dominant species in both environmental and respiratory specimens. In contrast to previous studies from the continental U.S., no *Mycobacterium avium* was identified. *Mycobacterium intracellulare* was found only in respiratory specimens and a soil sample. We conclude that Hawai'i's household water sources contain a unique composition of *Mycobacterium avium* complex (MAC), increasing our appreciation of NTM organisms of pulmonary importance in tropical environments.

Author Summary

In the U.S., the Hawaiian Islands have the highest number of nontuberculous mycobacterial (NTM) lung disease cases per capita. The tropical climate, geographical isolation of the islands, and aquifer water sources may have influence such prevalence. Previous studies suggest that NTM thrive in water biofilms and soil. To broaden our understanding of potential environmental reservoirs and species composition of NTM in the Hawaiian the Shoot for the Cure Foundation. NAH acknowledges the NIH Biomedical Informatics training grant 2T15LM009451-06. MS acknowledges the Boettcher Webb-Waring Foundation and the Potts Memorial Foundation. Finally, the University of Texas Health Science Center and National Jewish Health would like to acknowledge support from the Amon G. Carter Foundation. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist. Dr. Matthew J. Bankowski, Ph.D., D(ABMM) is the Technical Director for Diagnostic Laboratory Services, Inc., the reference laboratory for The Queen's Medical Center, which is owned by the Queen's Health Care system.

Islands, we sampled environmental sites and examined patient isolates. Our recovery and identification of *Mycobacterium chimaera* and several other clinically relevant NTM species and the absence of *Mycobacterium avium* in both the indigenous environment and clinical specimens underscore the need for further studies to define the environmental factors that drive NTM lung disease and species composition in high prevalence locations such as the Hawaiian Islands.

Introduction

Nontuberculous mycobacteria (NTM) are ubiquitous inhabitants of natural and human-engineered environments. To date, there are over 175 species of NTM with standing in nomenclature [1]. They range in virulence from benign environmental microorganisms to difficult-totreat human pathogens [2]. Potentially pathogenic NTM have been documented in households, institutions (*i.e.*, hospital premise plumbing), and soil [3]. In the continental United States (U. S.), household plumbing and environmental aerosols are thought to be important point sources of infection [4-8]. The most common NTM species to cause lung disease in the continental U. S. are those of the Mycobacterium avium complex (MAC)-slowly growing mycobacteria (SGM) that include Mycobacterium avium subsp. "hominissuis" and Mycobacterium intracellulare [9]. Clinically relevant environmental rapidly growing mycobacteria (RGM) include Mycobacterium abscessus subsp. abscessus, massiliense, and bolletii as well as the closely related species, Mycobacterium chelonae [10]. The current hypothesis is that NTM lung infections follow exposure to NTM from the home or other environmental source. [6]. Of interest, the predominant NTM species responsible for lung disease varies by geographic region, suggesting that environmental conditions (e.g., pH, oxygen, organic matter, and salinity) and the presence of other microorganisms influence NTM species numbers and diversity [11].

Despite the almost universal exposure to environmental NTM, pulmonary infections are relatively rare in otherwise healthy, non-bronchiectatic individuals and more common in individuals with abnormal lung architecture such as bronchiectasis and emphysema [12]. Never-theless, it is important to identify the environmental niches that harbor potentially pathogenic NTM in geographical areas with a high prevalence of disease. In the U.S., the Hawaiian Islands were found to have the highest period prevalence of NTM lung disease (396 cases/100,000 persons for a total ten year time period) in a sampling of 2.3 million Medicare Part B beneficiaries enrolled from 1997 to 2007 [13]. In a follow-up study, spatial modeling revealed high-prevalence locations for NTM lung disease in this state [14]. The Hawaiian Islands also showed the highest age-adjusted mortality rates from NTM lung disease in the U.S., particularly in women over 55 years of age [15].

The high prevalence of NTM lung disease in the Hawaiian Islands provided the impetus to explore potential sources of infection and to determine the predominating NTM species in both environmental and clinical specimens. These islands are recognized for their unique island geology, flora, and fauna which are largely impacted by the tropical climate and isolation of the archipelago in the Pacific Ocean. Unlike most areas in the continental U.S. for which surface water serves as the primary public water source, underground aquifers provide water there. The Hawaiian Islands are also home to the highest number of elderly Asian-Pacific Islanders in the U.S.—a population previously recognized to be more susceptible to NTM infection [14]. To better understand NTM lung disease as a neglected tropical disease of emerging importance in this geographic area, the objective of the current work was to employ state-of-the-art molecular techniques to describe the indigenous NTM species composition in

indoor and outdoor environments. A secondary objective was to analyze the genetic relatedness between the Hawaiian Island environmental NTM specimens (including 15 patient respiratory specimens) and continental U.S. NTM isolates.

Methods

Environmental sampling

In this cross-sectional study, we use the term "Hawaiian Islands" to designate the eight islands of the State of Hawai'i; the term "Hawai'i" refers to the youngest and largest island among the eight islands. Sample collection was conducted between December 2012 and January 2013. Samples were collected from 62 non-patient households located on the islands of Oahu, Molokai, Kauai, and Hawai'i. Detailed written instructions for collecting household water biofilms and soil samples were provided to local residents who volunteered to collect samples from their home as part of this study. As NTM are most commonly found in premise plumbing biofilms, samples were obtained by swabbing with sterile cotton-tipped applicators the inner surface of showerheads, kitchen and bath faucets, kitchen sink sprayers, refrigerator water dispensers, laundry room sinks, and shower drains [5, 6]. Samples from random sites in outdoor gardens or yards were also collected by clearing away surface leaves and other detritus and then scooping soil from the top five centimeters of ground into sterile 50 ml conical screw cap tubes as described [16].

Pilot samples of patient isolates

Respiratory isolates of slowly-growing NTM recovered from 15 de-identified Oahu patients suspected of mycobacterial lung disease whose sputum had been submitted for mycobacterial culture were randomly selected from saved isolates at Diagnostic Laboratory Services, Inc. (Aiea, HI). *Mycobacterium tuberculosis* was not recovered in any of these sputum samples where NTM were isolated. As these were de-identified patient residual isolates, where only age and gender were noted from routinely ordered laboratory testing, Institutional Review Board (IRB) consent was waived. However, it was impossible to determine whether these patients met current American Thoracic Society/Infectious Disease Society of America (ATS/IDSA) diagnostic criteria for NTM pulmonary disease as private health information were delinked [9].

Species and subspecies identification of NTM isolates by partial *rpoB* gene sequencing

Genome identification of environmental and patient NTM isolates was conducted through the amplification and sequencing of a 723 bp segment of the RNA polymerase beta subunit (*rpoB*) gene, also known as region 5 [17]. Sequences were trimmed for quality and compared against *rpoB* type strain sequences deposited in the National Center for Biotechnology Information (NCBI) GenBank using the BLAST algorithm. Definitions of species by single genes or spacer region were those of the Clinical Laboratory Standards Institute (CLSI) [18]. A sequence similarity cutoff of \geq 98.3% was used to determine the species identification according to previously described cutoffs validated by studies of rapidly-growing mycobacteria [17]. The sequencing of NTM strains derived from patients was approved by the National Jewish Health Human Subject IRB.

Non-Hawaiian Island NTM patient isolates

To determine whether NTM isolates from the Hawaiian Islands have shared sequence similarity with isolates obtained elsewhere, NTM type strains were included in genetic analyses. Type strains are denoted by superscript "T" and include *M. porcinum* CIP 105392 ^T, *M. abscessus* subsp. *abscessus* ATCC 19977^T, *M. abscessus* subsp. *bolletii* CIP 108541^T, *M. chelonae* ATCC 35752^T, and *M. chimaera* CIP 107892^T. Additionally, 33 clinical respiratory isolates of *M. chimaera* (one per patient) from seven other states–Maryland, Texas, Louisiana, North Carolina, Oregon, Mississippi, and Arkansas–submitted for molecular identification to the *Nocardia/Mycobacteria* Research Laboratory, University of Texas Health Science Center, Tyler, Texas were included. Those isolates were identified to species by partial 16S rRNA and region 5 *rpoB* gene sequencing. This work was approved by the Human Subjects Committee of the University of Texas Health Science Center, Tyler, Texas.

Nucleotide accession numbers

Partial *rpoB* gene sequences from 166 Hawaiian Island NTM isolates and 33 *M. chimaera* isolates from the continental U.S. were deposited in the GenBank nucleotide database. The GenBank accession numbers for type strain and representative isolate *rpoB* gene sequences of *M. porcinum*, *M. abscessus*, *M. chelonae*, and *M. chimaera* from NCBI are also listed in S1 Table

Phylogenetic and sequence variant network analyses

Partial *rpoB* sequences of respiratory and environmental NTM isolates (n = 166) were aligned using MUSCLE [19] and sequence alignments were trimmed to remove missing data from the ends of the final alignment. Phylogenetic trees were generated using the neighbor-joining method based on the number of nucleotide differences and uniform rates among sites while omitting any sites in the alignment with gaps or missing data in MEGA version 6 [20].

For *rpoB* sequence variant analyses, only sequences greater than 600bp and with no ambiguous base calls were included. Sequences were grouped by species and compared to selected type and non-type strain sequences from NCBI. The PopART population genetics software was used to examine intraspecies sequence variation, generate species-specific *rpoB* sequence variant networks, and label isolates by isolation source: *i.e.*, kitchen, bathroom, soil, patient [21]. For the *M. porcinum*, *M. abscessus*, and *M. chelonae* analyses, the environmental Hawaiian Island isolates and both type and non-type strains were included. For the *M. chimaera* analysis, environmental and clinical Hawaiian Island isolates, type, and non-type strains, as well as clinical isolates from seven states across the continental U.S. were included.

Statistical analyses

Statistical analyses were performed using R version 2.13.2 [22]. Fisher's Exact Tests were used to evaluate differences in proportions of NTM species or species groups between household areas (*i.e.*, bathroom, kitchen, and soil) or sample type (biofilm and soil).

Results

Environmental Areas Sampled

From a total of 62 households across four islands (Fig 1A), a total of 172 biofilm and soil samples were collected. The majority of the samples (n = 134, 78%) were collected from Oahu and included 35 showerheads (26%), 41 kitchen faucets (31%), 6 bathroom sink faucets (4%), 2 refrigerator water taps (1%), 3 other biofilm samples from laundry room faucets (2%), and 47 soil samples (35%). The remaining 38 samples (22%) were collected from 13 households on the neighbor islands.

Among all 172 biofilm and soil samples collected from the 62 households, NTM were isolated from 44% of samples (75/172) (Table 1). NTM were identified in nearly half of the samples on Oahu (65/134, 49%) and in approximately a quarter of samples from the neighbor



Fig 1. Environmental sampling for NTM. A) Biofilm swabs and soil samples were collected from 62 households on four of eight principal Hawaiian Islands. The numbers and proportions of total households from which NTM were recovered are shown. B) Locations of households sampled in towns across the island of Oahu. Colored triangles indicate sampling sites with biofilm or soil samples that were positive for presence of NTM (red) or negative for NTM (blue). C) The number of NTM species recovered from each household was calculated. Shown are the proportions of households harboring zero NTM species/household, one NTM species/household, two NTM species/household or three different NTM species/household.

doi:10.1371/journal.pntd.0005068.g001

	Oahu			Kauai			Molokai			Hawai'i n = 4		
	n = 49		n = 5		n = 4							
	# # %		# #	#	%	#	#	%	#	#	%	
	samples	NTM+	NTM+	samples	NTM+	NTM+	samples	NTM+	NTM+	samples	NTM+	NTM+
Showerhead	35	24	69%	5	4	80%	5	1	20%	2	0	0%
Kitchen	41	24	59%	3	2	67%	5	1	20%	3	1	33%
Bathroom	6	4	67%	ND	ND	ND	ND	ND	ND	2	0	0%
Refrigerator tap	2	2	100%	ND	ND	ND	ND	ND	ND	ND	ND	ND
Other	3	3	100%	1	1	100%	ND	ND	ND	ND	ND	ND
Total Biofilms	87	57	66%	9	7	78%	10	2	20%	7	1	14%
Soil	47	8	17%	3	0	0%	5	0	0%	4	0	0%
Total	134	65	49%	12	7	58%	15	2	13%	11	1	9%

Table 1. Description of household biofilm and yard/garden soil sampling and proportions of samples that were NTM culture-positive.

Total households sampled = 62

Total number of samples collected from households = 172 (n = 113 biofilm samples and n = 59 soil samples)

Total NTM culture positive samples = 75/172 (44%)

Other includes laundry and bedroom faucets

ND = Not done

doi:10.1371/journal.pntd.0005068.t001

islands (10/38, 26%). Overall, the NTM culture positivity rate for biofilms was 59% (67/113), which was significantly greater than for soil (14%, 8/59; $p = 6.0 \times 10^{-9}$).

NTM Recovered by Household

The majority of the environmental samples collected were from 49 households in seven different towns on Oahu, the most populated island (**Fig 1B**). NTM were recovered by culture from 82% of the Oahu households (**Fig 1A**). For the neighboring islands, NTM were also recovered in households on Kauai, Molokai, and Hawai'i (**Fig 1A**). Among the 62 collective households sampled in this study, only 14 had no NTM isolated (23%). However, the number of households with one, two, and three different NTM species isolated were 26/62 (42%), 18/62 (29%), and 4/62 (6%), respectively (**Fig 1C**). Overall, the majority of households (43/62, 69%) had at least one clinically relevant species of MAC, *M. abscessus* subsp., or *M. chelonae*—(**Table 2**).

NTM Recovery in Non-Household Samples

To determine the diversity of NTM in non-household sites, 13 environmental samples (n = 7 biofilm and n = 6 soil) were collected from eight public areas on Oahu and Kauai (Table 3). On Oahu, a total of six biofilms from public sites were collected including gymnasium shower-heads and water fountain taps. Four soil samples were also collected from public sites on Oahu. Two water biofilm and two soil samples were collected from public sites on Kauai. One Oahu public site soil sample contained *M. chimaera* (1/6 = 17%) and one biofilm sample contained *M. chelonae* (1/7 = 14%), but the majority (5/13 = 38%) yielded other RGM species (*i.e.*, *M. barrassiae*, *M. alvei*, and *M. septicum*). *RpoB* sequences from four distinct isolates did not have NCBI database matches above 95% sequence identity, suggesting they represent novel species.

Spectrum of NTM Species Identified from Environmental Samples

Among the 75 environmental samples from the households that were NTM culture-positive, 20 different NTM species were identified (Fig 2A) and 17% (13/75) grew out multiple NTM

NTM species	# of households	# of NTM species per household	High clinical prevalence NTM species*
None	14	0	N/A
<i>M. avium</i> complex (MAC)**	16	1	yes
M. abscessus or M. chelonae	5	1	yes
M. porcinum	3	1	no
Other NTM	2	1	no
M. abscessus or M. chelonae + other NTM	4	2	yes
MAC + M. abscessus or M. chelonae	6	2	yes
MAC + M. porcinum	4	2	yes
MAC + other NTM	4	2	yes
MAC + M. abscessus + other NTM	4	3	yes

Table 2. NTM species and combinations of species recovered from 62 households sampled from the Hawaiian Islands (includes biofilm and soil samples).

* NTM species associated with high clinical prevalence in previous epidemiological studies. Ref 10, 11.

** *M. avium* complex (MAC) includes *M. chimaera* and *M. intracellulare*. *M. intracellulare* was not recovered from biofilm samples, only a soil sample. *M. avium* was not recovered from any of the species examined.

doi:10.1371/journal.pntd.0005068.t002

species. The most common species recovered from households were MAC organisms with *M. chimaera* being the predominant species (42/75, 56%) (Fig 2B, left). The next most frequently isolated species were *M. chelonae* (12/75, 12%) and *M. porcinum* (11/75, 11%). All isolates of *M. abscessus* were confirmed as *M. abscessus* subsp. *abscessus* (10/75, 10%) [23, 24]. Less frequently isolated NTM species (<10%) included *M. phocaicum*, *M. gadium*, *M. alvei*, *M. gordonae*, *M. paraffinicum*, *M. marseillense*, and *M. colombiense*. No isolates of *M. avium* or *M. intracellulare* were recovered from household biofilm samples, though *M. intracellulare* was isolated from a

Table 3. Description and NTM recovery in non-household samples.

	Type of sample:	Source of sample:	NTM species identified:	Town, Island:	
1. a) Soil		a) Gym	a) None	Pearl City, Oahu	
	b) Biofilm swab	b) Hot tub (n = 1)	b) None		
2.	a) Soil	a) Fruit cannery	a) Fruit cannery a) <i>M. chimaera</i>		
	b) Biofilm swab	b) Water fountain (n = 1)	b) <i>M. gordonae</i>	-	
			c) Potential novel species *		
З.	a) Soil	a) Cemetery	a) <i>M. barrassiae</i>	Waipio, Oahu	
	b) Biofilm swab	b) Outdoor faucet (n = 1)	b) None		
4.	a) Soil	a) Gym [^]	a) Potential novel species*	Hawai'i Kai, Oahu	
	b) Biofilm swab	Biofilm swab b) Water fountain (n = 1) b) Potential no			
	c) Biofilm swab	c) Showerhead (n = 1)	c) None	-	
			d) None		
5.	Biofilm swab	Water fountain at a pier $(n = 1)$	M. gordonae	Honolulu, Oahu	
6.	Biofilm swab	Water fountain at tourist stop $(n = 1)$	M. chelonae	Waimea, Kauai	
7.	Soil	Tourist stop (n = 1)	a) <i>M. alvei</i>	Waimea, Kauai	
			b) Potential novel species***		
8.	Soil	Grotto (n = 1)	M. septicum	Kapa'a, Kauai	

* 93% identity to M. gadium

** 90% identity to M. jacuzzii

*** 94% identity to *M. alvei* and *M. fortuitum*.

[^] The only non-household site out of eight from which more than one household biofilm sample was collected.

doi:10.1371/journal.pntd.0005068.t003



Fig 2. Diversity and frequency of NTM species recovered from environmental samples. A) Phylogenetic analysis was performed from a multiple sequence alignment of partial rpoB sequences to illustrate the distribution of SGM and RGM isolates identified among environmental samples. Bolded names indicate NTM species in which more than one isolate was identified across the sample set. B) Proportions of samples positive for NTM species are shown for households (n = 75) and non-household sites (n = 9).

doi:10.1371/journal.pntd.0005068.g002



Fig 3. Household distribution of NTM species. Proportions of household samples harboring various NTM species are shown. Statistical comparisons between household locations were performed by Fisher's exact tests (*p = 0.05).

doi:10.1371/journal.pntd.0005068.g003

single soil sample. While *M. chimaera* and *M. chelonae* were identified in non-household samples, the majority classified as other NTM included potentially novel species (Fig 2B, right).

NTM Predominating in Household Locations

To determine whether NTM were present in particular household locations, the frequencies of NTM recovery between bathroom biofilms, kitchen biofilms, and soil were compared (Fig 3). *M. chimaera* was frequently identified from both bathroom (22/34, 65%) and kitchen (15/30, 50%) biofilms and was also identified in soil (2/7, 29%). *M. porcinum* was overrepresented in bathroom (8/34, 24%) compared to kitchen biofilms (2/30, 7%; p = 0.09), while *M. chelonae* was significantly more common in kitchen (9/30, 35%) compared to bathroom biofilms (3/34, 9%; *p = 0.05). *M. abscessus* was observed in similar proportions between bathroom (5/34, 15%) and kitchen (4/30, 13%) biofilms. *M. porcinum, M. chelonae*, and *M. abscessus* were not recovered from soil. *M. marseillense* was recovered only from soil and not identified in any of the household biofilm samples. NTM species that showed low prevalence in our study (*i.e.*, one isolate per species identified in the entire sample set and labeled "other RGM" and "other SGM") were primarily isolated from soil samples.

Genetic Diversity of Environmental RGM Isolates

To examine population diversity among RGM isolates from individual households, *rpoB* sequences of *M. porcinum*, *M. abscessus*, and *M. chelonae* were analyzed (Fig 4). Type and non-type strain *rpoB* sequences were included for comparison. In the *M. porcinum* dataset (n = 25





doi:10.1371/journal.pntd.0005068.g004

sequences), a total of seven sequence variants were identified (**Fig 4A**). All isolates from the bathroom, kitchen, and outside faucets were in the same sequence variant group as the *M. porcinum* type strain, CIP 105392^T, except for one kitchen isolate that contained a single SNP difference. Among all *M. abscessus* sequences (Hawaiian Island and type/reference strains; n = 38), six sequence variants of subsp. *abscessus*, four variants of subsp. *massiliense*, and one of subsp.*bolletii* (**Fig 4B**) were identified. Environmental *M. abscessus* isolates grouped with other *M. abscessus* subsp. *abscessus* and the majority of *M. abscessus* isolates (13/16 = 81%) shared an identical *rpoB* sequence with the type strain, ATCC 19977^T. Three additional isolates differed by one SNP each from the ATCC 19977^T type strain. Finally, *M. chelonae* isolates (**Fig 4C**) showed the greatest *rpoB* sequence variation with a total of 14 *rpoB* sequence variants. Hawaiian Island *M. chelonae* isolates fell into seven *rpoB* sequence variant groups, but the majority (15/20 = 80%) fell into two main subgroups: one group (6/15 and 40%) sharing the *M. chelonae* ATCC 19237 *rpoB* variant and a second group (5/15 and 33%) related to the *M. chelonae* ATCC 35752^T *rpoB* variant.

Genetic Diversity of Environmental and Clinical M. chimaera Isolates

As the majority of the Hawaiian Island environmental NTM isolates from this study were *M*. *chimaera*, 15 random respiratory SGM isolates from Oahu patients presenting to a pulmonary

	NTM Identified:	Age (yrs)	Gender
1.	M. chimaera	57	F
2.	M. chimaera	54	F
3.	M. chimaera	73	F
4.	M. chimaera	67	F
5.	M. chimaera	89	F
6.	M. chimaera	65	F
7.	M. chimaera	74	М
8.	M. chimaera	80	М
9.	M. chimaera	87	М
10.	M. chimaera	79	М
11	M. intracellulare	90	F
12.	M. intracellulare	67	F
13.	M. intracellulare	70	F
14.	M. intracellulare	87	F
15.	M. marseillense	78	М

Table 4. NTM species identified and demographic information of 15 pilot Oahu clinical isolates.

doi:10.1371/journal.pntd.0005068.t004

clinic with suspected mycobacterial lung disease were used as pilot samples to evaluate for the presence of *M. chimaera* in clinical specimens. As a group, the median age of the 15 patients was 75 years (95% CI, 68; 81 years) and 67% (10/15) were female (Table 4). Ten isolates were identified as *M. chimaera* (10/15, 67%), four as *M. intracellulare* (4/15, 27%), and one as *M. marseillense* (1/15, 6%). Of the ten patients with *M. chimaera*, 60% (6/10) were female. All four patients with *M. intracellulare* were female (100%; 5/5) and the patient with *M. marseillense* was male (Table 4). *M. avium* was not identified from any of the Oahu clinical isolates.

To measure the genetic similarity among a diverse collection of environmental and clinical *M. chimaera*, we analyzed *rpoB* sequence variation between the 57 Hawaiian Island environmental *M. chimaera* isolates and the 10 Oahu respiratory *M. chimaera* isolates. However, the *rpoB* sequence of one clinical *M. chimaera* isolate was excluded from these analyses due to the presence of ambiguous bases. Also included were NCBI non-type strains (n = 2), type strains (n = 2), and other *M. chimaera* respiratory isolates (n = 33) from seven states in the continental U.S. In total, 103 *M. chimaera* sequences were analyzed and only two *rpoB* sequence variants were observed (**Fig 5**). The larger variant subgroup comprised over 90% of the isolates including all of the Oahu respiratory and biofilm *M. chimaera* isolates. This group also contained the majority of continental U.S. clinical isolates and the CIP107892^T type strain. The smaller variant subgroup contained continental U.S. clinical isolates, non-type strains from NCBI, and Hawaiian Island soil isolates.

Discussion

To our knowledge, this is the first assessment of environmental NTM prevalence and species composition in the Hawaiian Islands. This archipelago is approximately halfway between the continental U.S. and Asia; thus, one might speculate that the spectrum of NTM observed mirrors the results from other environmental studies from the continental U.S. or Asia. Due to the prevalence of *M. avium* subsp. "hominissuis" reported in studies from the continental U.S. and Japan [25–27], we suspected this species would be prevalent in Hawaiian Island household biofilms and patient samples; however, it was seemingly absent, at least in the samples examined in this study. In general, NTM are rare in groundwater [29] whereas *M. avium* subsp. "hominissuis" has been isolated from surface water sources [28]. Aquifers provide most of the drinking water in the Hawaiian Islands [30] which may be one reason for the lack of *M. avium* detection



Fig 5. Distribution of *rpoB* sequence variants among Hawaiian Island environmental and clinical isolates of *M. chimaera* compared to type strains. A sequence variant network was created based on an alignment of partial *rpoB* gene sequences (n = 103 total sequences, out of 591 positions) including environmental and Oahu clinical isolates. Hash marks indicate SNP differences between adjacent isolate subgroups (circles). Isolates per *rpoB* variant (n = X) are specified for each observed variant. Type and non-

type strain sequences and random clinical isolates from the continental U.S. were included for comparison. The *M. chimaera* type strain is denoted by superscript "T".

doi:10.1371/journal.pntd.0005068.g005

in our samples. However, given the widespread prevalence of *M. chimaera* and the RGM in Hawaiian Island household biofilms, local aquifers may be a potential reservoir for *M. chimaera* and other NTM. Future studies are needed to examine this hypothesis.

To date, species diversity assessments of environmental NTM in other tropical Pacific Islands remains scant. A recent study described the identification of the *M. fortuitum* complex in Polynesian residents with suspected tuberculosis [31] and other reports from the area highlight NTM-associated skin disease [32, 33]. On Australia, *M. intracellulare* was reported as the species responsible for most lung disease cases and yet only *M. avium* subsp. "hominissuis", *M. kansasii*, and *M. abscessus* isolates had a species that match between patients and their house-hold water system [34, 35].

An unexpected finding of this study was the frequent identification of M. chimaera from both the environmental samples collected from bathroom, kitchen, and soil samples (Fig 3) and patient isolates with suspected mycobacterial lung disease. Although the number of patient isolates was small and their disease status were not known, the correspondence between the high proportion of both environmental and clinical M. chimaera isolates is intriguing and offers direction for future investigations. M. chimaera was first described in 2004 [36] and was recently reported to cause health-care associated infections after open-heart surgery with the use of heater-cooler units [37, 38]. As this is a relatively newly described species, there are no simple methods to differentiate M. chimaera from M. intracellulare. Furthermore, low frequency of presence in lung samples of patients from Germany, Italy, Zambia, and China [39-41] is most likely due to its misidentification as *M. intracellulare*. A greater adoption of more refined molecular methods to distinguish M. chimaera from M. intracellulare has facilitated the more precise speciation of M. chimaera (33). In a previous U.S. study, water biofilm isolates originally reported as M. intracellulare, proved to be M. chimaera or other MAC-X [4]. Provisionally, it appears that the main environmental source of M. chimaera in the Hawaiian Islands are water biofilms and less from the soil (Fig 3), whereas M. intracellulare was absent in water biofilms and only recovered from soil, consistent with the finding of others [4] (Fig 3, other SGM). Soil should also be regarded as a potential reservoir for *M. marseillense*.

Among our environmental samples, *M. porcinum*, *M. chelonae*, and *M. abscessus* were the most frequently identified RGM species. The *M. fortuitum* complex including *M. porcinum* were found to comprise the majority of clinical isolates examined in French Polynesia (42/87, 48%) using partial *rpoB* gene sequencing [31]. Of these, *M. porcinum* was identified in three patients who fulfilled ATS criteria for NTM lung disease. To our knowledge, *M. porcinum* infections have not yet been reported in the Hawaiian Islands, but the organism has been isolated from water supplies in other U.S. areas (*e.g.*, Texas) [42, 43]. *M. abscessus* was recently associated with an outbreak in cystic fibrosis patients at a hospital in Hawai'i [44]. *M. chelonae* infection was reported in a case study of an individual from Hawai'i after laser *in situ* keratomileusis (LASIK) surgery [45]. It is important to mention that among the environmental samples in this study, these particular RGM were more commonly identified in bathroom and kitchen biofilm samples and absent from soil (Fig 3), suggesting a preferential environmental niche for these particular RGM species.

Phylogenetic analyses were performed to evaluate whether the genetic diversity among environmental NTM species identified from the Hawaiian Island samples differed from those collected from the continental U.S. A relatively high genetic diversity among *M. chelonae* was observed with four major *rpoB* subgroups present, while most isolates of *M. porcinum* and *M*. *abscessus* belonged to one major genetic group per species (Fig 4). The presence of only two genetic subtypes of *M. chimaera* among a geographically diverse population of environmental and suspect respiratory Oahu specimens, as well as clinical isolates from seven other states in the continental U.S. suggests a low level of genetic divergence occurring in this species (Fig 5). Whole genome sequence comparisons will be necessary to improve our understanding of the genetic relationships between environmental and respiratory populations of *M. chimaera*.

This study has some limitations including the following in methodology: (i) we were unable to consistently collect a large number of samples from the same indoor sites for each participating household, (ii) a sampling bias exists as the majority of samples were collected from Oahu (home to the majority of the state's population) with only a few household samples collected from the less populated Molokai, Kauai, and Hawai'i and none from Kaho'olawe, Maui, Lanai, or Ni'ihau, and (iii) instead of a single person conducting all environmental sampling, household areas were sampled by local citizens, which added a layer of non-equivalency to the process of sample collection. To reduce non-uniformity in the collection process, we applied a well-accepted citizen science approach to minimize variability introduced by handling of samples by different people [46]. Although we cannot be certain our findings represent the true geographic diversity of NTM in the Hawaiian Islands, this work describes the largest study of environmental NTM in this geographic area with a documented high NTM disease burden. We would advocate for a larger, randomized systematic study of the distribution of environmental NTM in future work. To the best of our knowledge, all environmental samples were from households whose occupants are not known to have NTM lung disease; thus, it will be imperative to sample NTM patient households in a larger future study especially as a more thorough comparison of prevalence and numbers of NTM species in patients and their local environment can be assessed. We were also unable to confirm that the clinical isolates used in this study were etiological agents of respiratory disease or due to benign colonization from environmental exposures. Additionally, this pilot clinical isolate panel did not contain any RGM. Nevertheless, the observation that *M. chimaera* was the most common species in both environmental and clinical isolates examined suggests the possibility of environmental exposures and clinical NTM lung disease. To determine whether NTM in the household environment contributes to clinical disease, we hope to initiate a large-scale genomic study of matched household and clinical NTM isolates from NTM-infected Hawai'i patients who fulfill ATS/ IDSA criteria for lung disease. Undoubtedly, the data collectively presented in this study will be valuable in guiding the design of a more comprehensive study.

In summary, this study describes environmental sampling, microbiological selection, and molecular identification to determine the NTM species diversity in the Hawaiian Island environment. The observation that *M. chimaera* was the most common NTM species identified in both our Hawai'i environmental samples as well as in a small sampling of respiratory specimens from patients with suspected mycobacterial lung disease suggests that *M. chimaera* may be an important environmentally acquired respiratory pathogen. Furthermore, *M. chimaera* may be unique in prevalence in tropical climates such as Hawai'i. Additional studies with systematic collection of matched environmental and respiratory specimens, high-resolution genotyping methods, and correlation with demographic and epidemiological data (*i.e.* age, gender together with ethnicity and host risk and genetic factors) will be necessary to further characterize this observation and the important clinical implications.

Supporting Information

S1 Checklist. STROBE checklist. (DOC)

S1 Table. Accession Numbers For Study Isolates. *Mycobacteria* isolates derived from Hawaiian Island and Continental U.S.A. clinical and household specimens. Species identification and NCBI GenBank accession numbers are provided for each isolate. (PDF)

Acknowledgments

The authors thank the citizen scientists in Hawai'i for their enthusiastic participation (particularly Mrs. Colleen Takemori) and assistance with sample collection, and willingness to advance the science and medicine of mycobacterial translational research. Special appreciation is extended to John DeGroote, Director of the GeoTREE Center, University of Northern Iowa for his assistance in generating the NTM-household map used in this study.

Author Contributions

Conceptualization: JRH NAH RMD RJW EDC JOF MS.

Data curation: NAH RMD.

Formal analysis: NAH RMD.

Funding acquisition: JRH RJW EDC JOF MS.

Investigation: JRH NAH RMD MDW LEE PRR TS EI.

Methodology: JRH NAH RMD.

Project administration: JRH.

Resources: JRH NAH RMD MJB RJW EDC JOF MS.

Software: NAH RMD.

Visualization: JRH NAH RMD.

Writing - original draft: JRH NAH RMD LEE PRR MJB RJW EDC JOF MS.

Writing - review & editing: JRH NAH RMD LEE PRR MJB RJW EDC JOF MS.

References

- 1. Euseby J. List of prokaryotic names with standing in nomenclature 1997.
- 2. Falkinham JO 3rd. Ecology of nontuberculous mycobacteria-where do human infections come from? Semin Respir Crit Care Med. 2013; 34(1):95–102. doi: 10.1055/s-0033-1333568 PMID: 23460009
- **3.** Williams MM, Armbruster CR, Arduino MJ. Plumbing of hospital premises is a reservoir for opportunistically pathogenic microorganisms: a review. Biofouling. 2013; 29(2):147–62. doi: 10.1080/08927014. 2012.757308 PMID: 23327332
- 4. Wallace RJ Jr., lakhiaeva E, Williams MD, Brown-Elliott BA, Vasireddy S, Vasireddy R, et al. Absence of Mycobacterium intracellulare and the Presence of Mycobacterium chimaera in Household Water and Biofilm Samples of Patients in the U.S. With Mycobacterium avium Complex Respiratory Disease. J Clin Microbiol. 2013.
- Tichenor WS, Thurlow J, McNulty S, Brown-Elliott BA, Wallace RJ Jr., Falkinham JO 3rd. Nontuberculous Mycobacteria in household plumbing as possible cause of chronic rhinosinusitis. Emerg Infect Dis. 2012; 18(10):1612–7. doi: 10.3201/eid1810.120164 PMID: 23017381
- Falkinham JO 3rd. Nontuberculous mycobacteria from household plumbing of patients with nontuberculous mycobacteria disease. Emerg Infect Dis. 2011; 17(3):419–24. doi: 10.3201/eid1703.101510 PMID: 21392432

- De Groote MA, Pace NR, Fulton K, Falkinham JO. Relationships between *Mycobacterium* isolated from patients with pulmonary mycobacterial infection and potting soils. Appl Environ Microbiol. 2006; 72:7602–6. doi: 10.1128/AEM.00930-06 PMID: 17056679
- Feazel LM, Baumgartner LK, Peterson KL, Frank DN, Harris JK, Pace NR. Opportunistic pathogens enriched in showerhead biofilms. Proc Natl Acad Sci U S A. 2009; 106(38):16393–9. doi: 10.1073/ pnas.0908446106 PMID: 19805310
- Griffith DE, Aksamit T, Brown-Elliott BA, Catanzaro A, Daley C, Gordin F, et al. An official ATS/IDSA statement: diagnosis, treatment, and prevention of nontuberculous mycobacterial diseases. Am J Respir Crit Care Med. 2007; 175(4):367–416. doi: 10.1164/rccm.200604-571ST PMID: 17277290
- Thomson RM, Carter R, Tolson C, Coulter C, Huygens F, Hargreaves M. Factors associated with the isolation of Nontuberculous mycobacteria (NTM) from a large municipal water system in Brisbane, Australia. BMC Microbiol. 2013; 13:89. doi: 10.1186/1471-2180-13-89 PMID: 23601969
- Cook JL. Nontuberculous mycobacteria: opportunistic environmental pathogens for predisposed hosts. Br Med Bull. 2010; 96:45–59. doi: 10.1093/bmb/ldq035 PMID: 20977990
- Dirac MA, Horan KL, Doody DR, Meschke JS, Park DR, Jackson LA, et al. Environment or host?: A case-control study of risk factors for Mycobacterium avium complex lung disease. Am J Respir Crit Care Med. 2012; 186(7):684–91. doi: 10.1164/rccm.201205-0825OC PMID: 22859521
- Adjemian J, Olivier KN, Seitz AE, Holland SM, Prevots DR. Prevalence of nontuberculous mycobacterial lung disease in U.S. Medicare beneficiaries. Am J Respir Crit Care Med. 2012; 185(8):881–6. doi: 10.1164/rccm.201111-2016OC PMID: 22312016
- Adjemian J, Olivier KN, Seitz AE, Falkinham JO 3rd, Holland SM, Prevots DR. Spatial clusters of nontuberculous mycobacterial lung disease in the United States. Am J Respir Crit Care Med. 2012; 186 (6):553–8. doi: 10.1164/rccm.201205-0913OC PMID: 22773732
- Mirsaeidi M, Machado RF, Garcia JG, Schraufnagel DE. Nontuberculous mycobacterial disease mortality in the United States, 1999–2010: a population-based comparative study. PLoS One. 2014; 9(3).
- Brooks RW, Parker BC, Gruft H, Falkinham JO 3rd. Epidemiology of infection by nontuberculous mycobacteria. V. Numbers in eastern United States soils and correlation with soil characteristics. Am Rev Respir Dis. 1984; 130(4):630–3. doi: 10.1164/arrd.1984.130.4.630 PMID: 6486562
- Adekambi T, Colson P, Drancourt M. rpoB-based identification of nonpigmented and late-pigmenting rapidly growing mycobacteria. J Clin Microbiol. 2003; 41(12):5699–708. doi: <u>10.1128/JCM.41.12</u>. 5699-5708.2003 PMID: 14662964
- Institute CLS. Interpretive criteria for identification of bacteria and fungi by DNA target sequencing: guideline. Wayne, PA: 2008.
- Edgar RC. MUSCLE: multiple sequence alignment with high accuracy and high throughput. Nucleic acids research. 2004; 32(5):1792–7. doi: 10.1093/nar/gkh340 PMID: 15034147
- Tamura K, Stecher G, Peterson D, Filipski A, Kumar S. MEGA6: Molecular Evolutionary Genetics Analysis version 6.0. Molecular biology and evolution. 2013; 30(12):2725–9. doi: <u>10.1093/molbev/</u> mst197 PMID: 24132122
- Bandelt HJ, Forster P, Rohl A. Median-joining networks for inferring intraspecific phylogenies. Molecular biology and evolution. 1999; 16(1):37–48. PMID: 10331250
- 22. Team RDC. R: a language and environment for statistical computing. Vienna, Austria: R Foundation for Statistical Computing, 2011.
- Nash KA, Brown-Elliott BA, Wallace RJ Jr. A novel gene, erm(41), confers inducible macrolide resistance to clinical isolates of Mycobacterium abscessus but is absent from Mycobacterium chelonae. Antimicrob Agents Chemother. 2009; 53(4):1367–76. doi: 10.1128/AAC.01275-08 PMID: 19171799
- Brown-Elliott BA, Vasireddy S, Vasireddy R, Iakhiaeva E, Howard ST, Nash K, et al. Utility of sequencing the erm(41) gene in isolates of Mycobacterium abscessus subsp. abscessus with low and intermediate clarithromycin MICs. J Clin Microbiol. 2015; 53(4):1211–5. doi: 10.1128/JCM.02950-14 PMID: 25653399
- Griffith DE, Aksamit T, Brown-Elliott BA, Catanzaro A, Daley C, Gordin F, et al. An official ATS/IDSA statement: diagnosis, treatment, and prevention of nontuberculous mycobacterial diseases. Am J Respir Crit Care Med. 2007; 175:367–416. doi: 10.1164/rccm.200604-571ST PMID: 17277290
- Ito Y, Hirai T, Fujita K, Maekawa K, Niimi A, Ichiyama S, et al. Increasing patients with pulmonary Mycobacterium avium complex disease and associated underlying diseases in Japan. J Infect Chemother. 2015; 21(5):352–6. doi: 10.1016/j.jiac.2015.01.004 PMID: 25640532
- Ichikawa K, van Ingen J, Koh WJ, Wagner D, Salfinger M, Inagaki T, et al. Genetic diversity of clinical Mycobacterium avium subsp. hominissuis and Mycobacterium intracellulare isolates causing pulmonary diseases recovered from different geographical regions. Infect Genet Evol. 2015; 36:250–5. doi: 10.1016/j.meegid.2015.09.029 PMID: 26435330

- Falkinham JO, Norton CD, LeChevallier MW. Factors influencing numbers of *Mycobacterium avium*, *Mycobacterium intracellulare*, and other Mycobacteria in drinking water distribution systems. Appl Environ Microbiol. 2001; 67:1225–31. doi: 10.1128/AEM.67.3.1225-1231.2001 PMID: 11229914
- Martin EC, Parker BC, Falkinham JO 3rd. Epidemiology of infection by nontuberculous mycobacteria. VII. Absence of mycobacteria in southeastern groundwaters. Am Rev Respir Dis. 1987; 136(2):344–8. doi: 10.1164/ajrccm/136.2.344 PMID: 2441632
- Oki DS, Gingerich SB, Whitehead RL. Hawaii in Ground Water Atlas of the United States, Segment 13, Alaska, Hawaii, Puerto Rico and the US Virgin Islands US Geological Survey Hydrologic Investigations1999. p. N12-N22.
- Phelippeau M, Djaltou AO, Musso D, Drancourt M. Epidemiology of nontuberculous mycobacteria in French Polynesia. J Clin Microbiol. 2015.
- **32.** Lillis JV, Ansdell D. Outbreak of nontuberculous mycobacterial disease in the central Pacific. Dermatol Clin. 2011; 29(1):9–13. doi: 10.1016/j.det.2010.09.008 PMID: 21095522
- Lee MW, Brenan J. Mycobacterium marinum: chronic and extensive infections of the lower limbs in south Pacific islanders. The Australasian journal of dermatology. 1998; 39(3):173–6. PMID: 9737045
- Thomson RM. Changing epidemiology of pulmonary nontuberculous mycobacteria infections. Emerg Infect Dis. 2010; 16(10):1576–83. doi: 10.3201/eid1610.091201 PMID: 20875283
- Thomson R, Tolson C, Carter R, Coulter C, Huygens F, Hargreaves M. Isolation of nontuberculous mycobacteria (NTM) from household water and shower aerosols in patients with pulmonary disease caused by NTM. J Clin Microbiol. 2013; 51(9):3006–11. doi: 10.1128/JCM.00899-13 PMID: 23843489
- 36. Tortoli E, Rindi L, Garcia MJ, Chiaradonna P, Dei R, Garzelli C, et al. Proposal to elevate the genetic variant MAC-A, included in the Mycobacterium avium complex, to species rank as Mycobacterium chimaera sp. nov. Int J Syst Evol Microbiol. 2004; 54(Pt 4):1277–85. doi: 10.1099/ijs.0.02777-0 PMID: 15280303
- Sax H, Bloemberg G, Hasse B, Sommerstein R, Kohler P, Achermann Y, et al. Prolonged Outbreak of Mycobacterium chimaera Infection After Open-Chest Heart Surgery. Clin Infect Dis. 2015.
- Kohler P, Kuster SP, Bloemberg G, Schulthess B, Frank M, Tanner FC, et al. Healthcare-associated prosthetic heart valve, aortic vascular graft, and disseminated Mycobacterium chimaera infections subsequent to open heart surgery. Eur Heart J. 2015.
- Schweickert B, Goldenberg O, Richter E, Gobel UB, Petrich A, Buchholz P, et al. Occurrence and clinical relevance of Mycobacterium chimaera sp. nov., Germany. Emerg Infect Dis. 2008; 14(9):1443–6. doi: 10.3201/eid1409.071032 PMID: 18760016
- Malama S, Munyeme M, Mwanza S, Muma JB. Isolation and characterization of non tuberculous mycobacteria from humans and animals in Namwala District of Zambia. BMC research notes. 2014; 7:622. doi: 10.1186/1756-0500-7-622 PMID: 25204270
- Liu G, Chen ST, Yu X, Li YX, Ling Y, Dong LL, et al. Bacteriological and virulence study of a Mycobacterium chimaera isolate from a patient in China. Antonie Van Leeuwenhoek. 2015; 107(4):901–9. doi: 10.1007/s10482-015-0382-x PMID: 25763937
- Wallace RJ Jr., Brown-Elliott BA, Wilson RW, Mann L, Hall L, Zhang Y, et al. Clinical and laboratory features of Mycobacterium porcinum. J Clin Microbiol. 2004; 42(12):5689–97. doi: <u>10.1128/JCM.42</u>. 12.5689-5697.2004 PMID: 15583300
- Brown-Elliott BA, Wallace RJ Jr., Tichindelean C, Sarria JC, McNulty S, Vasireddy R, et al. Five-year outbreak of community- and hospital-acquired Mycobacterium porcinum infections related to public water supplies. J Clin Microbiol. 2011; 49(12):4231–8. doi: 10.1128/JCM.05122-11 PMID: 21998415
- 44. Johnston DI, Chisty Z, Gross JE, Park SY. Investigation of Mycobacterium abscessus outbreak among cystic fibrosis patients, Hawaii 2012. The Journal of hospital infection. 2016. Epub 2016/05/31.
- Rodriguez B, Holzinger KA, Le LH, Winkle RK, Allen RD. Mycobacterium chelonae keratitis after laserassisted subepithelial keratectomy. Journal of cataract and refractive surgery. 2006; 32(6):1059–61. doi: 10.1016/j.jcrs.2006.03.011 PMID: 16814072
- Abe J, Alop-Mabuti A, Burger P, Button J, Ellsberry M, Hitzeman J, et al. Comparing the temporal colonization and microbial diversity of showerhead biofilms in Hawai'i and Colorado. FEMS Microbiol Lett. 2016.